Author’s response to reviews

Title: A case of pseudoxanthoma elasticum with proliferative diabetic retinopathy

Authors:

Tsunehiko Ikeda (tikeda@osaka-med.ac.jp)
Keigo Kakurai (jangboohimalaya@gmail.com)
Maiko Hayashi (maimiosotide015@yahoo.co.jp)
Kanako Yamada (blue-blue1106@mail.goo.ne.jp)
Norihiko Ishizaki (ga78435@xa2.so-net.ne.jp)
Yumiko Yonemoto (choco_choco_cafe@yahoo.co.jp)
Seita Morishita (infinity_s2000@yahoo.co.jp)
Ryohsuke Kohmoto (ryousuke0218@hotmail.co.jp)
Takaki Sato (opt147@osaka-med.ac.jp)
Teruyo Kida (kida_teruyo@yahoo.co.jp)

Version: 2 Date: 10 Aug 2017

Author’s response to reviews:

Professor Hemant Khanna
Section Editor (Retina)
BMC Ophthalmology

BOPH-D-16-00065
A case of pseudoxanthoma elasticum with proliferative diabetic retinopathy

Dear Professor Khanna:
We greatly appreciate the thorough and constructive editor’s comment that we received on our submitted manuscript, entitled “A case of pseudoxanthoma elasticum with proliferative diabetic retinopathy”.

Please find the revisions that we have made to the manuscript.

We greatly thank you in advance for your valuable time and kind consideration, and we truly hope that you will find our revised manuscript now suitable for publication in BMC Ophthalmology.

Yours sincerely,

Tsunehiko Ikeda, M.D.
Professor and Chair
Department of Ophthalmology
Osaka Medical College
2-7 Daigaku-machi
Takatsuki-City, Osaka 569-8686, JAPAN
Tel: +81-72-684-6434; Fax: +81-72-682-0995
E-mail: tikeda@osaka-med.ac.jp